Author's response to reviews

Title: Antiviral efficacy and safety of abacavir-containing combination antiretroviral therapy as first-line treatment of HIV infected children and adolescents: a systematic review protocol.

Authors:

Olatunji O Adetokunboh (adetok_sic@yahoo.com)
Anel Schoonees (anelschoonees@sun.ac.za)
Charles S Wiysonge (charlesw@sun.ac.za)

Version: 2 Date: 25 July 2014

Author's response to reviews:

Olatunji O. Adetokunboh
Division of Community Health
Faculty of Medicine and Health Sciences
Stellenbosch University
Cape Town 7505
South Africa.
24 July 2014

The Editors-in-Chief,
Systematic Reviews.

Dear Editor,

On behalf of my co-authors, I have the honour to forward the corrections made with respect to the reviewer’s and editor’s comments concerning the submitted manuscript entitled, “Antiviral efficacy and safety of abacavir-containing combination antiretroviral therapy in the first-line treatment of HIV infected children and adolescents: a systematic review protocol.”

We have provided a point-by-point response to comments from the referee and editor in the attached table. The comments are reproduced in the second column and our responses in the third column.

We thank you for your kind consideration.

Yours sincerely,

Olatunji Adetokunboh